Author's response to reviews

Title: Overexpression p21WAF1/CIP1 in suppressing retinal pigment epithelial cells and progression of proliferative vitreoretinopathy via inhibition CDK2 and cyclin E

Authors:

Ying Wang (wangying_1026@yahoo.com)
Zhigang Yuan (yzq0158@163.com)
Caiyun You (youcaiyun@126.com)
Jindong Han (djindonghan@126.com)
Haiyan Li (lhyeye@163.com)
Zhuhong Zhang (zhangzhuhonghao@163.com)
Hua Yan (phuayan2000@163.com)

Version: 5 Date: 26 September 2014

Author's response to reviews: see over
Dear Editor,

Thank you editor for constructive comments and suggestions to this manuscript (MS: 3903436101358187). Some corrections or revisions were made according to your comments. We’ll try our best to answer or explain all questions you and reviewers have presented.
Below are our explanations on comments reviewer(s) presented.

Sincerely,
Ying Wang

Editorial Revisions:
1. Can you please clarify in your methods section, whether you received ethics committee approval and written informed consent for m volunteers for the use of peripheral blood from healthy individuals. If you did obtain ethics committee approval, please include the name of the ethics committee which approved your study.

Reply: I have added the statement of ethics committee approval in the methods (94-97).

2. Experimental research on vertebrates or any regulated invertebrates should have been approved by an appropriate animal ethics committee. A statement detailing compliance with guidelines and/or ethical approval must be included in the manuscript. Authors are encouraged to conform to the Animal Research: Reporting In Vivo Experiments (ARRIVE) guidelines (http://www.nc3rs.org.uk/page.asp?id=1357) for reporting animal studies.

Reply: I have added the statement of ethics committee approval in the methods (172-174).

3. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz.

Reply: Thanks for your suggestion and recommendation. I have copyedited the paper by Edanz.

I hope that the correction will meet with approval. Once again, thank you very much for your comments and suggestions.